Author's response to reviews

Title: Pulmonary fibrosis associated with psychotropic drug therapy: a case report

Authors:

Clare Thornton (clarethornton@doctors.org.uk)
Toby M Maher (t.maher@ucl.ac.uk)
David Hansell (d.hansell@rbht.nhs.uk)
Andrew G Nicholson (a.nicholson@rbht.nhs.uk)
Athol U Wells (a.wells@rbht.nhs.uk)

Version: 7 Date: 8 November 2009

Author's response to reviews:

Dear Editor

The changes made at the copy editing stage are, unfortunately, almost all grammatically incorrect. A number of the alterations made by the copy editor have resulted in the meaning of some sentences being lost or even changed e.g. the last sentence of the first paragraph of the discussion. Whilst I accept that there is a need to edit manuscripts so that they are in accord with 'house style', it is, to my mind, crucial that copy editors do not go beyond doing this and make changes that significantly alter the meaning of the original manuscript. I have therefore re-uploaded the previous version of the manuscript. My co-authors and I feel that this manuscript is grammatically correct.

Best wishes

Toby Maher